Author’s response to reviews

Title: Ginsenoside Rb1 Can Ameliorate the Key Inflammatory Cytokines TNF-α and IL-6 in a Cancer Cachexia Mouse Model

Authors:

Shuai Lu (lushuai805@163.com)
Yubo Zhang (zhangyb1995@163.com)
Huajun Li (lihuajun726@foxmail.com)
Jing Zhang (zhangjing0532@126.com)
Yingqian Ci (ciyingqian1127@163.com)
Mei Han (hanmei@bnu.edu.cn)

Version: 1 Date: 08 Jun 2019

Author’s response to reviews:

Dear Editor,

Thank you for your letter and comments concerning our manuscript entitled "Ginsenoside Rb1 Can Ameliorate the Key Inflammatory Cytokines TNF-α and IL-6 in a Cancer Cachexia Mouse Model" (BCAM-D-19-00781). We have studied comments carefully and have made correction which we hope meet with approval. Revised portion are marked in red in the paper.

The main corrections in the paper and the responds to the comments are as following:

1. In the third paragraph of the background section, we added the referenced evidence of the effect of ginsenoside Rb1 on cytokines TNF-α and IL-6 come from previous in animal experiments.

2. We have removed the Objective heading from abstract as the submission guidelines requested.

We appreciate for your warm work earnestly, and hope that the correction will meet with approval.

Once again, thank you very much for your comments and suggestion.

Mei Han